Author’s response to reviews

Title: Obesity prevention in child care: A review of U.S. state regulations

Authors:

Sara E Benjamin (sara_benjamin@harvardpilgrim.org)
Angie Cradock (acraddock@hsph.harvard.edu)
Elizabeth Walker (emwalker@NEMOURS.ORG)
Meghan Slining (slining@email.unc.edu)
Matthew W Gillman (matthew_gillman@harvardpilgrim.org)

Version: 2 Date: 29 April 2008

Author’s response to reviews:

Thank you for the thorough review of MS: 2290279981787845 - Obesity prevention in child care: A review of U.S. state regulations. As detailed below, we have addressed carefully each reviewer’s comments and made the appropriate changes to the text. We hope that the modifications to the manuscript have enhanced this paper.